Author’s response to reviews

Title: Prognostic value of TGF-β in lung cancer: systematic review and meta-analysis

Authors:

Cheng Shen (shencheng568_hx@163.com)
Jue Li (lijue16@163.com)
Xin Wang (wangxin_hx@yahoo.com)
Yutian Lai (laiyutian@163.com)
Kun Zhou (zhoukun_hx@yahoo.com)
Pengfei Li (lipengfei123@163.com)
Lunxu Liu (liulunxu@yahoo.com)
guowei che (cheguowei_hx@aliyun.com)

Version: 2 Date: 06 Nov 2018

Author’s response to reviews:

Dear Editor,

Many thanks for your letter about our manuscript and thanks for the referees taking time to deal with the manuscript and giving our suggestions. These suggestions are all valuable and very helpful for improving our paper, as well as the important guiding significance to our researches. According to the referees’ suggestions, we have carefully amended and corrected the related contents in our paper. In addition to the referees’ suggestions, we also have made some small revisions in the manuscript. These revisions will not influence the content and framework of the paper.